# Treatment in the Absence of Disease Reclassification Among Men on Active Surveillance for Prostate Cancer

Peter S. Kirk, MD, MS <sup>1</sup>, Kehao Zhu, MS<sup>2</sup>; Yingye Zheng, PhD<sup>2</sup>; Lisa F. Newcomb, PhD<sup>1,3</sup>; Jeannette M. Schenk, PhD, RD<sup>3</sup>; James D. Brooks, MD<sup>4</sup>; Peter R. Carroll, MD<sup>5</sup>; Atreya Dash, MD<sup>6</sup>; William J. Ellis, MD<sup>1</sup>; Christopher P. Filson, MD <sup>1</sup>, <sup>7</sup>; Martin E. Gleave, MD<sup>8</sup>; Michael Liss, MD<sup>9</sup>; Frances Martin, MD<sup>10</sup>; Jesse K. McKenney, MD<sup>11</sup>; Todd M. Morgan, MD<sup>12</sup>; Peter S. Nelson, MD<sup>13</sup>; Ian M. Thompson, MD <sup>11</sup>; Andrew A. Wagner, MD<sup>15</sup>; Daniel W. Lin, MD<sup>1,3</sup>; and John L. Gore, MD, MS<sup>1</sup>

**BACKGROUND:** Maintaining men on active surveillance for prostate cancer can be challenging. Although most men who eventually undergo treatment have experienced clinical progression, a smaller subset elects treatment in the absence of disease reclassification. This study sought to understand factors associated with treatment in a large, contemporary, prospective cohort. **METHODS:** This study identified 1789 men in the Canary Prostate Cancer Active Surveillance Study cohort enrolled as of 2020 with a median follow-up of 5.6 years. Clinical and demographic data as well as information on patient-reported quality of life and urinary symptoms were used in multivariable Cox proportional hazards regression models to identify factors associated with the time to treatment **RESULTS:** Within 4 years of their diagnosis, 33% of men (95% confidence interval [CI], 30%-35%) underwent treatment, and 10% (95% CI, 9%-12%) were treated in the absence of reclassification. The most significant factor associated with any treatment was an increasing Gleason grade group (adjusted hazard ratio [aHR], 14.5; 95% CI, 11.7-17.9). Urinary quality-of-life scores were associated with treatment without reclassification (aHR comparing "mostly dissatisfied/terrible" with "pleased/mixed," 2.65; 95% CI, 1.54-4.59). In a subset analysis (n = 692), married men, compared with single men, were more likely to undergo treatment in the absence of reclassification (aHR, 2.63; 95% CI, 1.04-6.66). **CONCLUSIONS:** A substantial number of men with prostate cancer undergo treatment in the absence of clinical changes in their cancers, and quality-of-life changes and marital status may be important factors in these decisions. **Cancer 2022;128:269-274**. © *2021 American Cancer Society.* 

#### LAY SUMMARY:

• This analysis of men on active surveillance for prostate cancer shows that approximately 1 in 10 men will decide to be treated within

- 4 years of their diagnosis even if their cancer is stable.
- These choices may be related in part to quality-or-life or spousal concerns.

**KEYWORDS:** active surveillance, prostatic neoplasms, quality of life.

## INTRODUCTION

Active surveillance is well recognized as an essential management approach for men with newly diagnosed clinically localized prostate cancer.<sup>1,2</sup> However, despite increasing adoption of this strategy, enrolling and maintaining men on active surveillance can be challenging for a multitude of reasons.<sup>3</sup> A number of clinical factors have been found to be associated with the risk of progression on active surveillance and the subsequent decision to pursue definitive therapy.<sup>4-14</sup>

Less well characterized are the reasons that men elect to discontinue active surveillance and pursue therapy for prostate cancer in the absence of clinical progression of their disease. The role of anxiety as an obstacle to both enrollment and retention on active surveillance protocols has been recognized.<sup>15-18</sup> However, it is unclear whether other significant factors may drive some men on active surveillance to pursue treatment without important changes in their cancer staging or grading. We hypothesized that there would be measurable differences in men's self-reported symptomatology and quality of life (QOL) if we compared patients undergoing treatment in the presence or absence of reclassification.

Additional supporting information may be found in the online version of this article.

DOI: 10.1002/cncr.33911, Received: July 14, 2021; Revised: August 6, 2021; Accepted: August 7, 2021, Published online September 13, 2021 in Wiley Online Library (wileyonlinelibrary.com)

Corresponding Author: Peter S. Kirk, MD, MS, Department of Urology, University of Washington, 1959 NE Pacific St, Seattle, WA 98195 (pskirk@uw.edu).

<sup>&</sup>lt;sup>1</sup>Department of Urology, University of Washington, Seattle, Washington; <sup>2</sup>Biostatistics Program, Public Health Sciences, Fred Hutchinson Cancer Research Center, Seattle, Washington; <sup>3</sup>Cancer Prevention Program, Public Health Sciences, Fred Hutchinson Cancer Research Center, Seattle, Washington; <sup>4</sup>Department of Urology, Stanford University, Stanford, California; <sup>5</sup>Department of Urology, University of California, San Francisco, California; <sup>6</sup>VA Puget Sound Health Care Systems, Seattle, Washington; <sup>7</sup>Department of Urology, Emory University, Atlanta, Georgia; <sup>8</sup>Department of Urologic Sciences, University of British Columbia, Vancouver, British Columbia, Canada; <sup>9</sup>Department of Urology, University of Texas Health Sciences Center, San Antonio, Texas; <sup>10</sup>Department of Urology, Eastern Virginia Medical School, Virginia Beach, Virginia; <sup>11</sup>Robert J. Tomsich Pathology and Laboratory Medicine Institute, Cleveland Clinic, Cleveland, Ohio; <sup>12</sup>Department of Urology, University of Michigan, Ann Arbor, Michigan; <sup>13</sup>Division of Human Biology and Clinical Research, Fred Hutchinson Cancer Research Center, Seattle, Washington; <sup>14</sup>CHRISTUS Medical Center Hospital, San Antonio, Texas; <sup>15</sup>Division of Urology, Beth Israel Deaconess Medical Center, Boston, Massachusetts

Specifically, we sought to identify factors associated with treatment in the absence of grade reclassification in a large, contemporary, prospective cohort of men on active surveillance for prostate cancer. Understanding the factors motivating these men to pursue treatment may inform the design of interventions to promote maintenance of active surveillance and thereby avoid or delay treatment and its associated side effects.

# MATERIALS AND METHODS

# Study Population

Data are from the Canary Prostate Active Surveillance Study (PASS), a prospective multicenter cohort initiated in 2008 of men diagnosed with clinically localized prostate cancer who choose active surveillance as their initial prostate cancer management strategy (clinicaltrials.gov NCT00756665), in which men are enrolled after providing informed consent under institutional review board supervision.<sup>19</sup> Under the PASS protocol, prostate-specific antigen (PSA) was measured every 3 months, clinic visits occurred every 6 months, and biopsies were performed 6 to 12 and 24 months after diagnosis and then every 2 years. Magnetic resonance imaging (MRI) was performed at the treating clinicians' discretion. The study population was 2003 men enrolled in PASS as of April 2020. For this analysis, men enrolled more than 5 years after their diagnosis (n = 62) or treated within the 6 months after their diagnosis or before the first surveillance biopsy (n = 46) were excluded. In addition, men who were reclassified to a higher biopsy Gleason grade group (GG) before enrollment (n = 90), were diagnosed with GG 3 (n = 8), or were missing data (n = 8) were also excluded; this left 1789 men in the analyses.

# Statistical Methods

The primary end points examined included 1) the time from diagnosis to any prostate cancer treatment and 2) the time from diagnosis to prostate cancer treatment without grade reclassification at a surveillance biopsy. For both end points, participants without treatment were censored at the date of last study contact. Patients who enrolled after their diagnosis were treated as left-truncated data and were considered at risk only after the time of enrollment. For models examining the time to treatment without reclassification, grade reclassification was treated as a competing event, and men who experienced an increase in grade were censored at the date of that event.

Overall cumulative incidences of both end points were estimated via the Aalen-Johansen estimator.

Time-varying covariate Cox proportional hazards regression models were used to examine the associations between the covariates of interest and the time to any treatment, and cause-specific Cox proportional hazards models were used for the time to treatment without a reclassification end point to account for competing risk. Information on clinical characteristics, pathologic characteristics, urinary symptoms, and demographics was considered for covariates, including GG at diagnosis, age at diagnosis, PSA at diagnosis (log-transformed), and race (Black, White, or other). Additionally, we considered increases in GG, the percentage of cores containing cancer, the prostate size (log-transformed), and differences in PSA from the diagnosis (log-transformed). The American Urological Association (AUA) urinary QOL score (delighted, pleased/mostly satisfied/mixed, or mostly dissatisfied/unhappy/terrible), the AUA symptom score, the body mass index, and the clinical T stage were modeled as time-varying covariates, with the most recent information before event times being used. Sensitivity analyses were conducted restricting the cohort to men with GG 1 at diagnosis (n = 1618) and altering the endpoint to time to treatment in the absence of grade or volume reclassification, where volume reclassification was defined as an increase in cancer volume from <34% to  $\ge34\%$  cores containing cancer (n = 131 events). An additional analysis was conducted with inclusion of available MRI results as a covariate. To explore whether marital status (married vs single) was associated with treatment overall and in the absence of reclassification, univariable and multivariable models were also evaluated among the subset of men enrolled after June 2015 (n = 692), when PASS started collecting these data. A 2-sided P value < .05 was considered statistically significant. Analyses were performed with R version 3.3.0.

# RESULTS

Table 1 summarizes the demographic and clinical characteristics of the study sample. The cohort was predominantly White men with GG 1, clinical stage T1 disease. The median AUA symptom score at enrollment was 7 (interquartile range, 3-12), and the most common AUA QOL scores were "pleased" and "mostly satisfied." With a median follow-up of 5.6 years (interquartile range, 2.5-8.6 years), 401 were treated after GG reclassification at biopsy, and 181 were treated without GG reclassification. Within 4 years of the diagnosis, the estimated cumulative incidence of being treated was 33% (95% confidence interval [CI], 30%-35%) and 10% (95% CI, 9%-12%)

	Not Treated (n = 1207)	Treated After Reclassification $(n = 401)$	Treated Without Reclassification $(n = 181)$	Total (n = 1789)
Age, y	63 (58-67)	63 (58-67)	63 (58-67)	63 (58-67)
Race				
White	1056 (87)	346 (86)	164 (91)	1566 (88)
Black	91 (8)	34 (8)	11 (6)	136 (8)
Other	60 (5)	21 (5)	6 (3)	87 (5)
BMI, kg/m <sup>2 a</sup>	27.6 (25.2-30.7)	27.7 (25.2-30.7)	26.6 (24.4-29.4)	27.5 (25.1-30.6
Gleason group				
1	1107 (92)	370 (92)	141 (78)	1618 (90)
2	100 (8)	31 (8)	40 (22)	171 (10)
PSA, ng/mL	5.1 (3.9-6.6)	5.1 (4.2-6.6)	5.0 (4.1-6.5)	5.1 (4-6.6)
Prostate size, cm <sup>3</sup>	46.5 (34.5-62.8)	35.5 (27.1-50.9)	39.6 (33.1-50.4)	43.4 (32.1-58.6
% positive cores	8.3 (8.3-16.7)	16.7 (8.3-25)	16.7 (8.3-25)	10 (8.3-16.7)
Clinical T stage <sup>a</sup>	. ,			. ,
T1	1083 (90)	352 (88)	160 (88)	1595 (89)
T2a	115 (10)	46 (11)	20 (11)	181 (10)
T2b/c	9 (1)	3 (1)	1 (1)	13 (1)
AUA symptom score <sup>a</sup>	7 (4-12)	7 (3-11)	6 (3-11)	7 (3-12)
AUA QOL <sup>a</sup>				
Delighted	266 (22)	107 (27)	56 (31)	429 (24)
Pleased	317 (26)	122 (30)	48 (27)	487 (27)
Mostly satisfied	347 (29)	97 (24)	43 (24)	487 (27)
Mixed	187 (15)	47 (12)	22 (12)	256 (14)
Mostly dissatisfied	56 (5)	19 (5)	7 (4)	82 (5)
Unhappy	25 (2)	4 (1)	4 (2)	33 (2)
Terrible	9 (1)	5 (1)	1 (1)	15 (1)
Years between diagno-	0.6 (0.3-1.2)	0.6 (0.3-0.9)	0.6 (0.3-1.3)	0.6 (0.3-1.1)
sis and enrollment				
Marital status <sup>a,b</sup>				
Married	429 (79)	81 (77)	38 (86)	548 (79)
Single	114 (21)	24 (23)	6 (14)	144 (21)

#### TABLE 1. Demographics and Clinical Data

Abbreviations: AUA, American Urological Association; BMI, body mass index; PSA, prostate-specific antigen; QOL, quality of life.

Numbers and percentages (in parentheses) are displayed for categorical variables, and medians and interquartile ranges (in parentheses) are displayed for continuous variables.

<sup>a</sup>At enrollment.

<sup>b</sup>Percentages are based on participants with a known marital status. Before 2015, marital status was not collected.

with and without grade reclassification, respectively. The subset of 692 men who were enrolled since 2015 when data about marital status were collected had similar characteristics (Supporting Table 1).

Table 2 highlights the univariable and multivariable associations of individual factors with the risk of any treatment during follow-up. The strongest factor independently associated with receiving treatment over time was GG upstaging, with an adjusted hazard ratio (aHR) of 14.5 for a 1-unit GG increase (95% CI, 11.7-17.9). Additional significant factors included GG at diagnosis, PSA, age, prostate size, volume of positive cores, and body mass index. The AUA QOL score was associated with the time to treatment (hazard ratio comparing "mostly dissatisfied/terrible" with "pleased/mixed," 1.72; 95% CI, 1.25-2.36; hazard ratio comparing "delighted" with "pleased/mixed," 1.31; 95% CI, 1.09-1.58; P < .001); however, the effects were attenuated and not significant after adjustments for other factors (P = .3). The AUA symptom score was not significantly associated with any treatment.

Table 3 displays the univariable and multivariable associations of individual factors with the time to treatment in the absence of GG reclassification. Similarly to the associations for any treatment, GG at diagnosis, PSA, and the percentage of positive cores were all strongly associated with treatment without reclassification. The AUA QOL score was significantly associated with treatment without reclassification (aHR comparing "mostly dissatisfied/terrible" with "pleased/mixed," 2.65; 95% CI, 1.54-4.59). Among the subset of 692 men with marital status data, marital status was also independently associated with treatment in the absence of reclassification (aHR for married men vs single men, 2.63; 95% CI, 1.04-6.66), although it was not associated with any treatment (Supporting Tables 2 and 3). Associations of AUA QOL with treatment in the absence of reclassification were similar when the analysis was restricted to men with GG 1 at diagnosis (aHR, 3.45; 95% CI, 1.9-6.38) and when the end point excluded volume reclassification (aHR, 2.41; 95% CI, 1.28-4.53). Overall, 593 men (33%) had MRI imaging performed at some point

	-			
	Univariate Hazard Ratio	Р	Multivariate Hazard Ratio	Ρ
Gleason grade		<.001		<.001
group <sup>a</sup>				
1 (referent)	1.00		1.00	
2	1.83 (1.42-2.34)		1.73 (1.33-2.26)	
Increase in		<.001		<.001
Gleason grade				
group				
1	25.5 (21.1-30.8)		14.5 (11.7-17.9)	
2	64.67 (49.6-84.4)		36.6 (27.4-49.0)	
3	84.1 (54.2-130.7)		31.1 (19.2-50.2)	
4	140.0 (69.8-280.9)		52.3 (25.3-108.4)	
PSA <sup>a,b</sup>	1.31 (1.14-1.51)	<.001	1.68 (1.39-2.02)	<.001
Difference in PSA <sup>b</sup>	2.63 (2.23-3.1)	<.001	1.7 (1.44-2.01)	<.001
Age <sup>a</sup>	1.00 (0.99-1.01)	.7	0.98 (0.97-0.99)	.003
Prostate size <sup>b</sup>	0.46 (0.39-0.56)	<.001	0.70 (0.56-0.89)	.003
% positive cores <sup>c</sup>	1.83 (1.77-1.89)	<.001	1.31 (1.24-1.37)	<.001
Clinical T stage		<.001		.13
T1c (referent)	1.00		1.00	
T2a	1.96 (1.55-2.49)		1.2 (0.93-1.55)	
T2b+	2.07 (1.19-3.6)		0.68 (0.38-1.2)	
AUA symptom	0.95 (0.89-1.02)	.17	1.07 (0.99-1.16)	.11
score <sup>d</sup>				
AUA QOL score		<.001		.3
Delighted	1.31 (1.09-1.58)		1.12 (0.9-1.38)	
Pleased/	1.00		1.00	
mixed (referent)				
Mostly dis-	1.72 (1.25-2.36)		1.21 (0.85-1.72)	
satisfied/	· · · · ·		( , , , , , , , , , , , , , , , , , , ,	
terrible				
Race		.2		.4
White	1.00		1.00	
(referent)				
Black	1.36 (1-1.84)		1.20 (0.88-1.65)	
Other	1.03 (0.70-1.52)		0.86 (0.58-1.28)	
BMI	1.00 (0.98-1.02)	>.9	0.98 (0.96-1)	.029

**TABLE 2.** Univariable and Multivariable Results of a Cox Proportional Hazards Model of Any Treatment Among PASS Participants

Abbreviations: AUA, American Urological Association; BMI, body mass index; PASS, Prostate Active Surveillance Study; PSA, prostate-specific antigen; QOL, quality of life.

<sup>a</sup>At diagnosis.

<sup>b</sup>Log.

<sup>c</sup>Per 10% increase.

<sup>d</sup>Per 5-unit increase.

during their enrollment in PASS. The inclusion of MRI as a covariate did not appreciably alter the other associations, and although the MRI results were significant in the model for any treatment (aHR for Prostate Imaging– Reporting and Data System score  $\geq 4$  vs no MRI, 1.52; 95% CI, 1.20-1.91) they were not significant in the model of treatment without reclassification (aHR, 1.59; 95% CI, 0.99-2.56). Sensitivity analyses using PSA density rather than PSA and prostate volume in the models did not show any appreciable differences in the other reported associations. **TABLE 3.** Univariable and Multivariable Results of a Cox Proportional Hazards Model Predicting Treatment in the Absence of Grade Reclassification

	Univariate Hazard Ratio	Р	Multivariate Hazard Ratio	Р
Gleason		<.001		.011
grade				
group <sup>a</sup>				
1 (referent)	1.00		1.00	
2	3.62 (2.54-5.16)		1.69 (1.14-2.51)	
PSA <sup>a,b</sup>	1.32 (1.02-1.7)	.029	1.79 (1.28-2.52)	<.001
Difference in PSA <sup>b</sup>	2.48 (1.84-3.34)	<.001	2.31 (1.67-3.18)	<.001
Age <sup>a,b</sup>	0.99 (0.97-1.02)	.6	0.98 (0.96-1)	.093
Prostate size <sup>b</sup>	0.59 (0.43-0.81)	.001	0.69 (0.46-1.04)	.076
% positive cores <sup>c</sup>	1.84 (1.71-1.97)	<.001	1.7 (1.57-1.84)	<.001
Clinical T stage		.043		.9
T1c (referent)	1.00		1.00	
T2a	1.79 (1.15-2.8)		1.15 (0.71-1.85)	
T2b+	1.76 (0.56-5.51)		1.05 (0.33-3.36)	
AUA symp- tom score <sup>d</sup>	1.00 (0.89-1.12)	>.9	1.05 (0.91-1.21)	.5
AUA QOL		<.001		.002
score Delighted	1.47 (1.06-2.05)		1.26 (0.87-1.84)	
Pleased/ mixed (referent)	1.00		1.00	
Mostly dis- satisfied/ terrible	2.78 (1.71-4.54)		2.65 (1.54-4.59)	
Race		.7		.5
White (referent)	1.00		1.00	
Black	1.01 (0.55-1.85)		1.02 (0.55-1.89)	
Other	0.73 (0.32-1.64)		0.62 (0.27-1.42)	
BMI	0.97 (0.94-1.01)	.093	0.98 (0.95-1.01)	.3

Abbreviations: AUA, American Urological Association; BMI, body mass index; PSA, prostate-specific antigen; QOL, quality of life.

<sup>a</sup>At diagnosis.

<sup>b</sup>Log.

<sup>c</sup>Per 10% increase.

<sup>d</sup>Per 5-unit increase.

# DISCUSSION

We found that within this large, contemporary cohort of men on active surveillance for prostate cancer, approximately one-third underwent treatment within 4 years of their diagnosis, and 1 in 10 pursued treatment in the absence of grade reclassification. For most patients, GG upgrading is the strongest predictor of prostate cancer treatment. However, for men who end up being treated without GG reclassification, urinary QOL, independently of clinical factors, appears to be an important factor. Exploratory analyses also suggest that married men are more likely to undergo treatment without reclassification. Our findings suggest that factors other than cancer characteristics play an important role in men on active surveillance electing to pursue treatment for their prostate cancer. It should be noted, however, that disease characteristics remain important predictors of treatment even in the absence of grade reclassification. This may reflect both a lower threshold to pursue treatment in patients with more aggressive baseline characteristics (GG 2 disease, more positive cores, and/or higher PSA levels at diagnosis) and changes such as rising PSA levels that prompt treatment before these changes are manifested in the form of grade reclassification.

The results of this study are consistent with prior work examining the decision to pursue treatment rather than continuing active surveillance.<sup>13,20</sup> Longitudinal, population-based data have suggested that as many as 20% of men electing to discontinue active surveillance do so because of personal preference rather than biopsy or PSA progression.<sup>21</sup> Prior results from the PASS cohort showed that within the first 2 years of follow-up on active surveillance, nearly one-third of men who decided to pursue treatment did so in the absence of adverse disease reclassification.<sup>22</sup> This work helps to clarify the motivations and factors involved in the decision-making for those men.

Interestingly, although QOL was significantly associated with treatment without reclassification in this study, the symptom score was not, and this could imply that other unmeasured factors may be driving the lower observed QOL. One possible explanation is that patients with worsened anxiety surrounding their diagnosis and active surveillance interpret their symptoms differently and with more severe detriment to their QOL even in the absence of measurable differences in their urinary symptoms. Cancerrelated anxiety was previously associated with worsened urinary symptoms.<sup>23</sup> The results of our sensitivity analyses suggest that although the symptom score is associated with treatment without reclassification, this relationship is not independent of QOL. Furthermore, the lack of an attenuation of the association between QOL and treatment without reclassification between univariable and multivariable models further implies additional unmeasured factors affecting QOL. Work assessing factors influencing QOL scores has found that additional psychological factors such as anxiety and depression are independent factors influencing urinary QOL score results.<sup>24</sup>

We also found in an exploratory subset analysis that marital status is an independent factor associated with the pursuit of treatment in the absence of reclassification. This could be from the anxiety of the partner influencing decision-making. Alternatively, spouses may affect the way in which patients perceive their symptoms and QOL and help to define what is tolerable versus intolerable. Either way, this result suggests that any intervention to help to optimize the retention of men enrolled in active surveillance will need to incorporate partners in the decision-making process.

This study has several important limitations. Although this work includes data from surveys of urinary symptoms and QOL, it will be important for future work in this area to include comprehensive measures of anxiety that we did not have available for these analyses. There has been significant study of anxiety as a driving factor in the decision of men not to pursue active surveillance.<sup>15,16</sup> Our study is further limited by the intrinsic selection of the men included in this study. They opted to enroll in a prospective cohort study, and their motivations for pursuing treatment could theoretically differ from those of community urology patients. However, if there were a difference, we would expect that it would favor even larger differences than those measured here because resources and support to help to manage the stress of active surveillance are likely more available to study participants than general patients with prostate cancer in the community. Lastly, this data source does not include detailed information regarding the treating physicians, and it is likely that treating physicians' preferences and biases may play some role in the decision-making regarding the discontinuation of active surveillance.

Nonetheless, this study has important implications for patients undergoing active surveillance for prostate cancer and for the providers caring for these men. Those attempting to retain men on active surveillance in the absence of disease progression will need to find ways to target and improve QOL, whether by directly managing lower urinary tract symptoms medically or surgically or by better controlling cancer-related and general anxiety. These approaches will also need to incorporate the partners and caregivers of patients to fully address this issue.

In summary, we demonstrate that an important subgroup of men on active surveillance pursue treatment in the absence of clinically important changes in their cancers and that lower reported urinary QOL and marital status appear to be important factors related to these decisions in addition to prostate cancer characteristics. These insights could be used to design focused interventions to help to prevent premature treatment and potentially avoidable side effects. Similarly, spouses and caregivers should be incorporated as key stakeholders in shared decision-making and decision aid–based discussions. Doing so could enable increased, longer term participation of more men on active surveillance for prostate cancer.

## FUNDING SUPPORT

This study was funded by the National Institutes of Health (U01 CA224255). Christopher P. Filson reports support from the American Cancer Society (MRSG-18-1-CPHPS) and the Winship Cancer Institute (Prostate Cancer Pilot Grant). Peter S. Nelson reports support from the National Cancer Institute (P50CA097186).

## CONFLICT OF INTEREST DISCLOSURES

Peter R. Carroll reports support from Myriad (to his institution). Todd M. Morgan reports payments from Blue Earth and the *European Urology* editorial board. The other authors made no disclosures.

## AUTHOR CONTRIBUTIONS

Peter S. Kirk: Conceptualization, visualization, and writing-original draft. Kehao Zhu: Conceptualization, data curation, formal analysis, software, visualization, and writing-original draft. Yingye Zheng: Formal analysis, methodology, and writing-review and editing. Lisa F. Newcomb: Conceptualization, methodology, project administration, supervision, and writing-review and editing. Jeanette M. Schenk: Methodology and writing-review and editing. James D. Brooks: Conceptualization and writing-review and editing. Peter R. Carroll: Conceptualization and writing-review and editing. Atreya Dash: Conceptualization and writingreview and editing. William J. Ellis: Conceptualization and writing-review and editing. Christopher P. Filson: Conceptualization and writing-review and editing. Martin E. Gleave: Conceptualization and writing-review and editing. Michael Liss: Conceptualization and writing-review and editing. Frances Martin: Conceptualization and writing-review and editing. Jesse K. McKenney: Conceptualization and writing-review and editing. Todd M. Morgan: Conceptualization and writing-review and editing. Peter S. Nelson: Conceptualization and writing-review and editing. Ian M. Thompson: Conceptualization and writing-review and editing. Andrew A. Wagner: Conceptualization and writing-review and editing. Daniel W. Lin: Conceptualization, funding acquisition, methodology, supervision, and writing-review and editing. John L. Gore: Conceptualization, methodology, supervision, and writing-original draft.

### REFERENCES

- Sanda MG, Cadeddu JA, Kirkby E, et al. Clinically localized prostate cancer: AUA/ASTRO/SUO guideline. Part I: risk stratification, shared decision making, and care options. *J Urol.* 2018;199:683-690. doi:10.1016/j.juro.2017.11.095
- Sanda MG, Cadeddu JA, Kirkby E, et al. Clinically localized prostate cancer: AUA/ASTRO/SUO guideline. Part II: recommended approaches and details of specific care options. *J Urol.* 2018;199:990-997. doi:10.1016/j.juro.2018.01.002
- Bokhorst LP, Alberts AR, Rannikko A, et al. Compliance rates with the Prostate Cancer Research International Active Surveillance (PRIAS) protocol and disease reclassification in noncompliers. *Eur Urol.* 2015;68:814-821. doi:10.1016/j.eururo.2015.06.012
- Welty CJ, Cowan JE, Nguyen H, et al. Extended followup and risk factors for disease reclassification in a large active surveillance cohort for localized prostate cancer. *J Urol.* 2015;193:807-811. doi:10.1016/j. juro.2014.09.094
- Matta R, Hird AE, Dvorani E, et al. Rates of primary and secondary treatments for patients on active surveillance for localized prostate cancer—a population-based cohort study. *Cancer Med.* 2020;9:6946-6953. doi:10.1002/cam4.3341
- Kinsella N, Stattin P, Cahill D, et al. Factors influencing men's choice of and adherence to active surveillance for low-risk prostate cancer: a mixed-method systematic review [figure presented]. *Eur Urol.* 2018;74:261-280. doi:10.1016/j.eururo.2018.02.026

- Kearns JT, Faino AV, Newcomb LF, et al. Role of surveillance biopsy with no cancer as a prognostic marker for reclassification: results from the canary Prostate Active Surveillance Study [formula presented]. *Eur* Urol. 2018;73:706-712. doi:10.1016/j.eururo.2018.01.016
- Iremashvili V, Manoharan M, Rosenberg DL, Soloway MS. Biopsy features associated with prostate cancer progression in active surveillance patients: comparison of three statistical models. *BJU Int.* 2013;111:574-579. doi:10.1111/j.1464-410X.2012.11127.x
- Drost FJH, Nieboer D, Morgan TM, et al. Predicting biopsy outcomes during active surveillance for prostate cancer: external validation of the Canary Prostate Active Surveillance Study risk calculators in five large active surveillance cohorts. *Eur Urol.* 2019;76:693-702. doi:10.1016/j. eururo.2019.07.041
- Cooperberg MR, Brooks JD, Faino AV, et al. Refined analysis of prostate-specific antigen kinetics to predict prostate cancer active surveillance outcomes. *Eur Urol.* 2018;74:211-217. doi:10.1016/j. eururo.2018.01.017
- Newcomb LF, Zheng Y, Faino AV, et al. Performance of PCA3 and TMPRSS2:ERG urinary biomarkers in prediction of biopsy outcome in the Canary Prostate Active Surveillance Study (PASS). *Prostate Cancer Prostatic Dis.* 2019;22:438-445. doi:10.1038/s41391-018-0124-z
- Schenk JM, Newcomb LF, Zheng Y, et al. African American race is not associated with risk of reclassification during active surveillance: results from the Canary Prostate Cancer Active Surveillance Study. J Urol. 2020;203:727-733. doi:10.1097/JU.000000000000621
- Simpkin AJ, Tilling K, Martin RM, et al. Systematic review and metaanalysis of factors determining change to radical treatment in active surveillance for localized prostate cancer. *Eur Urol.* 2015;67:993-1005. doi:10.1016/j.eururo.2015.01.004
- Van Hemelrijck M, Ji X, Helleman J, et al. Reasons for discontinuing active surveillance: assessment of 21 centres in 12 countries in the Movember GAP3 Consortium. *Eur Urol.* 2019;75:523-531. doi:10.1016/j.eururo.2018.10.025
- Latini DM, Hart SL, Knight SJ, et al. The relationship between anxiety and time to treatment for patients with prostate cancer on surveillance. *J Urol.* 2007;178:826-832. doi:10.1016/j.juro.2007.05.039
- Marzouk K, Assel M, Ehdaie B, Vickers A. Long-term cancer specific anxiety in men undergoing active surveillance of prostate cancer: findings from a large prospective cohort. J Urol. 2018;200:1250-1255. doi:10.1016/j.juro.2018.06.013
- Seaman AT, Taylor KL, Davis K, et al. Why men with a low-risk prostate cancer select and stay on active surveillance: a qualitative study. *PLoS One.* 2019;14:e0225134. doi:10.1371/journal.pone.0225134
- Wade J, Donovan J, Lane A, et al. Strategies adopted by men to deal with uncertainty and anxiety when following an active surveillance/ monitoring protocol for localised prostate cancer and implications for care: a longitudinal qualitative study embedded within the ProtecT trial. *BMJ Open.* 2020;10:e036024. doi:10.1136/bmjopen-2019-036024
- Newcomb LF, Brooks JD, Carroll PR, et al. Canary Prostate Active Surveillance Study: design of a multi-institutional active surveillance cohort and biorepository. *Urology*. 2010;75:407-413. doi:10.1016/j. urology.2009.05.050
- Bokhorst LP, Valdagni R, Rannikko A, et al. A decade of active surveillance in the PRIAS study: an update and evaluation of the criteria used to recommend a switch to active treatment. *Eur Urol.* 2016;70:954-960. doi:10.1016/j.eururo.2016.06.007
- Loeb S, Folkvaljon Y, Makarov DV, Bratt O, Bill-Axelson A, Stattin P. Five-year nationwide follow-up study of active surveillance for prostate cancer. *Eur Urol.* 2015;67:233-238. doi:10.1016/j.eururo.2014.06.010
- Newcomb LF, Thompson IM, Boyer HD, et al. Outcomes of active surveillance for clinically localized prostate cancer in the prospective, multi-institutional Canary PASS cohort. J Urol. 2016;195:313-320. doi:10.1016/j.juro.2015.08.087
- Tan HJ, Marks LS, Hoyt MA, et al. The relationship between intolerance of uncertainty and anxiety in men on active surveillance for prostate cancer. *J Urol.* 2016;195:1724-1730. doi:10.1016/j. juro.2016.01.108
- 24. Choi WS, Heo NJ, Lee YJ, Son H. Factors that influence lower urinary tract symptom (LUTS)–related quality of life (QoL) in a healthy population. *World J Urol.* 2017;35:1783-1789. doi:10.1007/s0034 5-017-2052-2